Published Ahead of Print on August 3, 2017, as doi:10.3324/haematol.2017.171355. Copyright 2017 Ferrata Storti Foundation.



Italian real life experience with brentuximab vedotin: results of a large observational study on 40 relapsed/refractory systemic anaplastic large cell lymphoma

by Alessandro Broccoli, Cinzia Pellegrini, Alice Di Rocco, Benedetta Puccini, Caterina Patti, Guido Gini, Donato Mannina, Monica Tani, Chiara Rusconi, Alessandra Romano, Anna Vanazzi, Barbara Botto, Carmelo Carlo-Stella, Stefan Hohaus, Pellegrino Musto, Patrizio Mazza, Stefano Molica, Paolo Corradini, Angelo Fama, Francesco Gaudio, Michele Merli, Angela Gravetti, Giuseppe Gritti, Annalisa Arcari, Patrizia Tosi, Anna Marina Liberati, Antonello Pinto, Vincenzo Pavone, Filippo Gherlinzoni, Virginia Naso, Stefano Volpetti, Livio Trentin, Maria Cecilia Goldaniga, Maurizio Bonfichi, Amalia De Renzo, Corrado Schiavotto, Michele Spina, Sergio Storti, Angelo Michele Carella, Vittorio Stefoni, Lisa Argnani, and Pier Luigi Zinzani

Haematologica 2017 [Epub ahead of print]

Citation: Broccoli A, Pellegrini C, Di Rocco A, Puccini B, Patti C, Gini G, Mannina D, Tani M, Rusconi C, Romano A, Vanazzi A, Botto B, Carlo-Stella C, Hohaus S, Musto P, Mazza P, Molica S, Corradini P, Fama A, Gaudio F, Merli M, Gravetti A, Gritti G, Arcari A, Tosi P, Liberati AM, Pinto A, Pavone V, Gherlinzoni F, Naso V, Volpetti S, Trentin L, Goldaniga MC, Bonfichi M, De Renzo A, Schiavotto C, Spina M, Storti S, Carella AM, Stefoni V, Argnani L, and Zinzani PL. Italian real life experience with brentuximab vedotin: results of a large observational study on 40 relapsed/refractory systemic anaplastic large cell lymphoma. Haematologica. 2017; 102:xxx doi:10.3324/haematol.2017.171355

Publisher's Disclaimer.

E-publishing ahead of print is increasingly important for the rapid dissemination of science. Haematologica is, therefore, E-publishing PDF files of an early version of manuscripts that have completed a regular peer review and have been accepted for publication. E-publishing of this PDF file has been approved by the authors. After having E-published Ahead of Print, manuscripts will then undergo technical and English editing, typesetting, proof correction and be presented for the authors' final approval; the final version of the manuscript will then appear in print on a regular issue of the journal. All legal disclaimers that apply to the journal also pertain to this production process.

Original Article: Italian real life experience with brentuximab vedotin: results of a large observational study on 40 relapsed/refractory systemic anaplastic large cell lymphoma

Alessandro Broccoli,^{1*} Cinzia Pellegrini,^{1*} Alice Di Rocco,² Benedetta Puccini,³ Caterina Patti,⁴ Guido Gini,⁵ Donato Mannina,⁶ Monica Tani,⁷ Chiara Rusconi,⁸ Alessandra Romano,⁹ Anna Vanazzi,¹⁰ Barbara Botto,¹¹ Carmelo Carlo-Stella,¹² Stefan Hohaus,¹³ Pellegrino Musto,¹⁴ Patrizio Mazza,¹⁵ Stefano Molica,¹⁶ Paolo Corradini,¹⁷ Angelo Fama,¹⁸ Francesco Gaudio,¹⁹ Michele Merli,²⁰ Angela Gravetti,²¹ Giuseppe Gritti,²² Annalisa Arcari,²³ Patrizia Tosi,²⁴ Anna Marina Liberati,²⁵ Antonello Pinto,²⁶ Vincenzo Pavone,²⁷ Filippo Gherlinzoni,²⁸ Virginia Naso,²⁹ Stefano Volpetti,³⁰ Livio Trentin,³¹ Maria Cecilia Goldaniga,³² Maurizio Bonfichi,³³ Amalia De Renzo,³⁴ Corrado Schiavotto,³⁵ Michele Spina,³⁶ Sergio Storti,³⁷ Angelo Michele Carella,³⁸ Vittorio Stefoni,¹ Lisa Argnani¹ and Pier Luigi Zinzani¹

¹Institute of Hematology "L. e A. Seràgnoli", University of Bologna, Bologna, Italy; ²Hematology, Department of Cellular Biotechnologies and Hematology, Sapienza University, Rome, Italy;

³Hematology Department, University and Hospital Careggi, Firenze, Italy;

⁴Dept. of Hematology, Azienda Ospedali Riuniti Villa Sofia Cervello, Palermo, Italy ⁵Department of Hematology, Ospedali Riuniti di Ancona, Ancona, Italy;

⁶Hematology Unit, Ospedale Papardo, Messina, Italy;

⁷Hematology Unit, Santa Maria delle Croci Hospital, Ravenna, Italy;

⁸Division of Hematology, Niguarda Hospital, Milan, Italy;

⁹Division of Hematology, AOU, Catania, Italy;

¹⁰European Institute of Oncology, Milano, Italy;

- ¹¹SC Ematologia, Azienda Ospedaliera Universitaria Città della Salute e della Scienza, Torino, Italy;
- ¹²Oncology and Hematology, Humanitas Cancer Center, Humanitas Clinical and Research Center, Rozzano, Italy;
- ¹³Institute of Hematology, Catholic University, Rome, Italy;
- ¹⁴ Scientific Direction, IRCCS-CROB, Referral Cancer Center of Basilicata, Rionero in Vulture (Pz);
- ¹⁵Ospedale Moscati Dept. of Hematology-Oncology, Taranto, Italy;
- ¹⁶Department of Hematology, Ciaccio-Pugliese Hospital, Catanzaro, Italy;
- ¹⁷Department of Hematology, Fondazione IRCCS Istituto Nazionale dei Tumori, University of Milan, Milan, Italy;
- ¹⁸Hematology Unit, Arcispedale Santa Maria Nuova, Istituto di Ricovero e Cura a Carattere Scientifico, Reggio Emilia, Italy;
- ¹⁹Hematology Unit, Policlinico di Bari, Bari, Italy;
- ²⁰Hematology, Ospedale di Circolo, Fondazione Macchi, Varese, Italy;
- ²¹Division of Hematology and Stem Cell Transplantation Unit, Cardarelli Hospital, Napoli, Italy;
- ²²Department of Hematology, Hospital Papa Giovanni XXIII, Bergamo, Italy;
- ²³Division of Hematology, Guglielmo da Saliceto Hospital, Piacenza, Italy;
- ²⁴Hematology Unit, Infermi Hospital Rimini, Rimini, Italy;
- ²⁵Hematology, Ospedale Perugia, Perugia, Italy;
- ²⁶Hematology-Oncology and Stem Cell Transplantation Unit, National Cancer Institute, Fondazione Pascale, IRCCS, Napoli, Italy;
- ²⁷Division of Hematology, Ospedale G. Panico, Tricase, Lecce, Italy;
- ²⁸Hematology Unit, Ca' Foncello Hospital, Treviso, Italy;
- ²⁹Sant'Andrea Hospital Sapienza, Roma, Italy;

³⁰Department of Hematology, DISM, Azienda Sanitaria Universitaria Integrata, Udine, Italy;

³¹Hematology and Clinical Immunology Unit, Department of Medicine, University of Padua,

Padua, Italy;

³²OncoHematology Unit, Fondazione Ca' Granda IRCCS Ospedale Maggiore Policlinico,

Milan, Italy;

³³Hematology, IRCCS Policlinico San Matteo, Pavia, Italy;

³⁴Hematology, AOU Federico II Napoli, Napoli, Italy;

³⁵Hematology, San Bortolo Hospital, Vicenza, Italy;

³⁶Division of Medical Oncology A, National Cancer Institute, Aviano, Italy;

³⁷Hematology, Università Cattolica Sacro Cuore Campobasso, Campobasso, Italy;

³⁸Division of Hematology 1, IRCCS A.O.U. San Martino IST, Genova, Italy

*AB and CP contributed equally to this work.

Running head: Brentuximab vedotin in ALCL: real life experience

Corresponding Author

Pier Luigi Zinzani

Institute of Hematology "L. e A. Seràgnoli"

University of Bologna

Via Massarenti, 9 – 40138 Bologna, Italy

Phone: +39 051 214 3680

Fax: +39 051 214 4037

E-mail: pierluigi.zinzani@unibo.it

Abstract word count: 247.

Text word count: 2484.

Tables: 1.

Figures: 3.

Supplemental file: 0.

Key Words: brentuximab vedotin, long-term response, real life, anaplastic large cell lymphoma, stem cell transplantation

ABSTRACT

From November 2012 to July 2014, brentuximab vedotin was available in Italy for patients with relapsed systemic anaplastic large cell lymphoma outside a clinical trial context according to the national law 648/96. A large Italian observational retrospective study was conducted on the use of brentuximab vedotin in the everyday clinical practice to check if clinical trial results are confirmed even in a real life context. Primary endpoint was the best response; secondary endpoints were the overall response rate at the end of the treatment, duration of response, survival and the safety profile. A total of 40 heavily pretreated patients were enrolled. Best response was observed after a median of 4 cycles in 77.5%: globally, 47.5% patients obtained a complete response, 64.2% in the elderly subset. Overall response rate was 62.5%. At the latest follow up 15/18 patients are still in complete response (3 with consolidative procedure). Progression free survival was 39.1% at 24 months and disease free survival 54% at 24 months (median not reached). Particularly, all the long term responders were aged <30 years at first infusion. The treatment was well tolerated even in this real life context and no death has been linked to drug toxicity. Brentuximab vedotin induces clinical responses quite rapidly, i.e. within the first 4 cycles in most responder patients, thus permitting the timely application of the transplantation phase. For patients ineligible for transplant or for who transplant failed, brentuximab vedotin may represent a feasible effective therapeutic option in everyday clinical practice.

Introduction

Approximately 40% to 65% of patients with systemic anaplastic large-cell lymphoma (sALCL) develop recurrent disease after front-line therapy. Historically, at relapse the disease is resistant to conventional multiagent chemotherapy regimens, and there is no established standard of care. High-dose therapy and autologous stem cell transplantation (ASCT) may result in long-term remission in 30% to 40% of patients, but the benefit is limited to patients with chemotherapy-sensitive disease. ²⁻⁶

Given that most patients with relapsed and refractory (R/R) sALCL are scheduled to undergo a highly toxic high-dose chemotherapy regimen, any strategy aimed at achieving a minimal disease status, specifically a positron emission tomography (PET)-negative status before ASCT, without severe toxicity would represent a major advance in the overall management of these patients. Furthermore, although the role of ASCT, the outcomes remain poor in those with primary chemorefractory disease, where long-term survival rarely exceeds 15-17%. In fact, disease recurrence still remains the principal cause of ASCT failure, and early disease progression after transplant, i.e. within 6 months from high-dose conditioning, emerges as the most important predictor of unfavorable outcome. No standard treatment options exist for patients showing disease relapse after ASCT or for patients not eligible for ASCT. In fact, while allogeneic stem cell transplantation (alloSCT) may induce a long-term progression-free survival (PFS) in a fraction of patients, only a few are candidates for this procedure, mainly as a result of unsatisfactory pre-transplant cytoreduction and substantial risk of morbidity due to the heavy load of previous therapies. Under this light, the optimization of the outcomes obtained with high-dose regimens and ASCT still remains a current strategic priority, in order to offer the best chance of cure for the largest fraction of patients with R/R disease.

Brentuximab vedotin (BV) is an antibody-drug conjugates targeting CD30 which may represent the optimal candidate among the new developed agents for the treatment of R/R sALCL.⁷ In fact, sALCL is characterized by the expression of CD30. In the initial phase 1 study of BV in patients with CD30+ lymphoid diseases, both the 2 sALCL patients achieved a complete response (CR).⁷ The favorable activity of this agent in R/R sALCL was clearly documented by Pro et al in a phase 2 study involving 58 patients: 86% obtained a response, which was a CR in 57% of cases.⁸ Median PFS for these patients was 13.3 months, and the median overall survival (OS) was not reached (estimation is 64% at 4 years). The same relevant proportion of CR in this subset of patients also emerges from the data collected by Zinzani et al regarding the BV Named Patient Program (NPP) experiences across Europe.^{9,11,15-17}

This high response rate is important not only in pretreated patients showing a poor prognosis, but also in first-line R/R patients because a CR obtained before transplant is one of the stronger predictors for long-term survival. BV can represent an optimal therapeutic option as a bridge to both ASCT and alloSCT program in patients achieving a suboptimal response after salvage treatment. Recent updates on the pivotal study have shown that BV can induce long lasting CR in sALCL pretreated cases either without additional consolidation therapies, suggesting that BV may be curative for some patients. The pooled overall response rate (ORR) and CR rate reported for R/R sALCL patients (globally 46) in NPP cohorts were 69.5% for both rates. 9,11,15-17

After accelerated approval by US Food and Drug Administration, eligible patients in Italy were granted early access through a NPP. After the closure of NPP, between 2012 BV was available in Italy for patients with R/R sALCL, based on a local disposition of the Italian Drug Agency (AIFA) issued according to a national law (Law 648/96: "medicinal products that are provided free of charge on the national health service"): a boundary zone

in the passage from clinical trials to marketing and free use phases where patients can be treated in any case.

On the basis of our previous explorative study, ¹⁸ a large Italian observational retrospective study was conducted on the use of BV in R/R sALCL patients in the everyday clinical practice to check if clinical trial results are confirmed even in a real life context.

Methods

An observational retrospective study was conducted among patients with sALCL treated from November 2012 to July 2014 with BV in 38 Italian centers outside of clinical trials, according to the national law no. 648/96.¹⁹ The study was approved by our institutional board (Azienda Ospedaliera di Bologna, Policlinico S.Orsola-Malpighi, coordinating center) and by all involved Ethical Committees and registered in the Italian Registry of Observational Studies. All participants gave written informed consent in accordance with the Declaration of Helsinki. A shared database was used after the approval of all the co-investigators and variables were strictly defined to avoid bias in reporting data.¹⁹ We obtained a special permission (for scientific purpose) from our Ethical Committee to collect even data of patients who were deceased or lost to follow up.

BV is administered as a 30-minute infusion of at the dose of 1.8 mg/kg of body weight every 3 weeks for a maximum of 16 cycles. Dose reduction to 1.2 mg/kg is recommended in case of grade 3 toxicity and the treatment has to be interrupted in case of grade 4 toxicity.

The primary endpoint of the study was the best response achieved during BV therapy; secondary endpoints were the ORR at the end of the treatment, duration of response (DoR), OS, PFS, disease free survival (DFS), and the drug safety and tolerability. Duration of therapy was defined as the number of cycle performed.

Effectiveness was also evaluated through the occurrence of long term responder (LTR) patients, defined as patients who have response (CR or partial response [PR]) duration ≥12 months. Response is assessed by PET/CT scan after cycle 4, 8, 12 and at drug discontinuation by each investigator using the International Working Group revised response criteria for malignant lymphoma.²⁰ Safety and tolerability were evaluated by recording incidence, severity, and type of any adverse event (AE) according to the National Cancer Institute Common Terminology Criteria for AEs v4.0.

OS was defined as the time from initiation of therapy to death from any cause and was censored at the date of the last available follow up. PFS was measured from initiation of therapy to progression, relapse, or death from any cause and was censored at the date of the last available follow up. DFS was calculated for CR patients from the first documentation of response to the date of relapse or death due to lymphoma or acute toxicity of treatment. DoR was calculated from the first objective tumor response (CR or PR) to first documentation of progression or death.²⁰ Lost to follow up patients (N=2). were censored at the latest available date.

Demographics and patients' characteristics as well as AEs were summarized by descriptive statistics. Survival functions were estimated by using the Kaplan-Meier method and were compared using log-rank test. Statistical analyses were performed with Stata 11 (StataCorp LP, TX) and p values were set at 0.05.

Results

Of the estimated 40 patients who received BV under the Law 648/96, all participated in this observational study. All had histologically confirmed CD30+ disease. The median age at BV was 47 years (range, 17-80 years) with 14 (35.0%) elderly (age> 60 years) patients; 28 were males and 12 were females. Eleven (27.5%) had systemic symptoms at baseline (Table 1).

The median number of prior lymphoma-related systemic regimens was 2 (range, 2-10) including high dose chemotherapy and autologous stem cell transplantation (ASCT) (in 13, 32.5% of the patients). Eight patients (20%) had received prior radiation therapy. Eighteen were anaplastic lymphoma kinase negative (ALK+) and 22 anaplastic lymphoma kinase positive (ALK-) status. For each patient the status after both frontline therapy and most recent therapy was collected: 24 (60%) patients had disease that was refractory to frontline therapy and 25 patients (62.5%) had disease that was refractory to last therapy before BV.

Response to treatment

Best response was observed after a median of 4 cycles in 31 (77.5%) patients: 19 (47.5%) obtained a CR and 12 (30%) achieved a partial response (PR). Overall responses rate at the end of the treatment was 62.5% (25 patients) represented by 18 (45%) CR and 7 (17.5%) PR; among the remaining patients, one had stable disease (SD), and 14 patients showed progression of disease (PD), respectively.

The best response rate was higher in the elderly subset (>60 years): 9 (64.3%) CR and 3 (21.4%) PR with an overall response rate of 85.7%. Four patients who were in CR at first restaging relapsed during further BV courses; 2 patients who were in PR at first restaging converted to CR status after the 4 subsequent infusions.

All patients who were in SD or PD at first restaging did not improve their status at the end of therapy. The median number of cycles administered was 8 (range 1-16). With a median follow up of 18 months global OS was 56.9% at 24 months (Figure 1) and median not reached yet. PFS at 24 months was 39.1%, median achieved at 12.5 months (Figure 2). DFS was 54% at 24 months (Figure 3); 4 out of 19 (21%) CR patients relapsed and 15 patients were in continuous CR (CCR) at the last follow up with a median DoR of 12 months (range, 9-24 months). After controlling for confounding variables, no differences in any time to point endpoints between ALK- and ALK+ patients were observed.

Among the CRs, 3 patients had consolidation with transplant (1 ASCT and 2 alloSCT). Currently, 15 patients are still in CR and, in particular, the 3 with consolidative procedure; among elderly patients, 6 out of 9 (66.7%) patients are still in CCR without any consolidative procedure after a median of 14 months. There are 5 LTR patients and all of them are still in CCR at the last available follow up. To note, they were all aged <35 years at BV therapy and only 1 of them had a subsequent consolidative transplant. At the latest follow-up, 27 (67.5%) patients were alive and 13 deceased (11 due to lymphoma, 2 for complications after alloSCT: one for respiratory insufficiency related to graft-versus-host-disease and 1 for pneumonia).

Safety

All patients who received at least one BV infusion were included in the safety analysis. In general, the treatment was well tolerated and the toxicity profile was very similar to the previously published data. Twelve patients had at least one toxicity. All hematologic toxicities but one were grade 1-2. In fact, we registered a grade 3 neutropenia. The extra-hematologic side effects were mostly represented by peripheral sensorial neurological toxicity (15/20) and, among them, 3 were grade 3. The other AEs were by nausea grade 1-2 (two patients), erythema grade 2 (two patients), hyposthenia grade 2 (one patient). Neurological toxicity always reversed completely after end of treatment. No long-term toxicity related to BV was observed during the follow-up period, even in patients later subjected to transplant consolidation.

Discussion

This retrospective large multicenter Italian study on 40 patients with R/R sALCL treated with BV outside a clinical trial represents the largest ever reported in a real world context. Our results are in accordance to the pivotal phase II study and its updates and to

the other national experience studies with an ORR of 77.5% and a CR rate of 47.5% in terms of best response.^{8,9,11,14-18}

In addition, we extrapolate some interesting consideration about the role of BV in everyday clinical practice. First, both the best response rate and ORR were higher in the elderly setting: 85.7% vs 77.5% and 64.3% vs 62.5%, respectively.

To be in CR after 4 cycles is confirmed very important for classifying the patient as a real good responder; at the same time, the right number of cycles to be performed for evaluating the potential consolidation with transplant (in the major part allogeneic transplant) or the continuation with BV until the cycle 16 remains an open issue, mainly because in case of CR the choice between the two options is at the physician discretion. According to the recent update by Pro et al. on the pivotal phase II study, the 5-year PFS was 68% in CR patients submitted to alloSCT versus 47% in patients who continued BV treatment even though they had obtained CR after the first 4 cycles.^{8,14} In this update the authors reported that 27.6% of the whole study population has achieved long-term remission exceeding 5 years in response to single agent BV without any additional anticancer therapy, other than transplant. In our study the estimated DFS at 2 years was 54% and 15 patients (37.5%) are in CCR with a median DoR of 12 months (range 9-24 months). As only 3/15 patients had transplant consolidation, comparison between them and patients who did not received SCT procedure was not possible. Thus, also in the real life experience as in the in the pivotal study, DoR and DFS indicate that, among R/R sALCL, a substantial subset of patients who obtained CR with single agent BV either obtained a long-term disease control and may potentially be cured. An important question remains unclear: among the patients in CR, which may benefit from the transplant consolidation? In our series there are 5 LTR patients and all of them are still in CCR at the latest follow-up and, in particular, only 1 with a consolidative alloSCT procedure. Update on pivotal study and our data could indicate that it is possible to obtain long disease

control also without transplant consolidation with the real chance to cure a subset of R/R sALCL only with BV.¹⁴ Physicians are still divided on whether or not offer a consolidative transplant to CR patients as solid clinical trial data are lacking on this issue. A large, well-designed randomized control study is needed, but ALCL is so rare that we are unlikely to ever have a definitive answer.

Differences in survival outcomes between ALK+ and ALK- patients have been often reported: no statistical significance in our sample was observed between the two subgroups.²¹

Our study indicated that for patients who obtained a SD or PD after 4 cycles of BV the potential conversion rate to PR or CR with further administrations is close to zero. The final message is that when patients show SD or PD at first restaging, they have to be shifted rapidly to another treatment. On the other hand, for patients who achieved PR after first restaging it could be important to continue the treatment: in our series 2/12 (16.7%) patients showed a conversion from PR to CR status.

In conclusion, the results of this large retrospective study on 40 R/R sALCL in the daily practice support the efficacy of single agent BV with manageable toxicity without evidence of cumulative toxic effects with previous regimens. We acknowledge that this kind of reports carry potential bias as the lacking of predictable and calculated sample size and the risk of toxicity underreporting. ALCL represents approximately 2% to 3% of all lymphoid neoplasms, thus it is a very rare disease. The phase II study who lead to an FDA accelerated approval of BV enrolled globally 58 patients, thus 40 ALCL patients from a single nation is a substantial sample related to this pathology. However, we could not analyze prognostic features due to the small sample and we reported the raw observed data. Observational studies may better identify clinically important AEs when compared with randomized controlled trials, for several reasons. Those reasons include longer follow-up times, the inclusion of patients with concomitant illnesses who may be more

likely to experience drug interactions or other side effects and to detect infrequent or rare complications. Regarding the retrospective nature of this specific study, AIFA has a strict monitoring on drugs prescribed under the law 648/96 and physicians have to report any AE occurring during treatment: thus, all the safety data are already in the patients' chart at the time our retrospective study starts.

In particular, our report confirms the activity in elderly patients, the duration of the clinical response independently by the transplant consolidation, and the relevance of the CR status after 4 cycles in term of final response. BV is the first drug which led to a drastic management change in ALCL, entailing an ORR of 80%. Next research efforts could be aimed at developing combination regimens with BV to reach the 100% of response in R/R ALCL patients.

References

- Savage KJ, Harris NL, Vose JM, et al. ALK- anaplastic large-cell lymphoma is clinically and immunophenotypically different from both ALK+ ALCL and peripheral T-cell lymphoma, not otherwise specified: Report from the International Peripheral T-Cell Lymphoma Project. Blood. 2008;111(12):5496-5504.
- Kewalramani T, Zelenetz AD, Teruya- Feldstein J, et al. Autologous transplantation for relapsed or primary refractory peripheral T-cell lymphoma. Br J Haematol. 2006;134(2):202-207.
- 3. Vose JM, Peterson C, Bierman PJ, et al. Comparison of high-dose therapy and autologous bone marrow transplantation for T-cell and B-cell non- Hodgkin's lymphomas. Blood. 1990(2);76:424-431.
- 4. Rodriguez J, Munsell M, Yazji S, et al. Impact of high-dose chemotherapy on peripheral T-cell lymphomas. J Clin Oncol. 2001(17);19:3766-3770.
- Moskowitz CH, Nimer SD, Glassman JR, et al. The International Prognostic Index predicts for outcome following autologous stem cell transplantation in patients with relapsed and primary refractory intermediate-grade lymphoma. Bone Marrow Transplant. 1999;23(6):561-567.
- Shipp MA, Abeloff MD, Antman KH, et al. International Consensus Conference on High- Dose Therapy with Hematopoietic Stem Cell Transplantation in Aggressive Non-Hodgkin's Lymphomas: Report of the jury. J Clin Oncol 1999;17(1):423-429.
- Younes A, Bartlett NL, Leonard JP, et al. Brentuximab vedotin (SGN-35) for relapsed CD30-positive lymphomas. N Engl J Med. 2010;363(19):1812-1821.

- Pro B, Advani R, Brice P, et al. Brentuximab vedotin (SGN-35) in patients with relapsed or refractory systemic anaplastic large-cell lymphoma: results of a phase II study. J Clin Oncol. 2012;30(18):2190-2196.
- Zinzani PL, Sasse S, Radford J, Shonukan O, Bonthapally V. Experience of brentuximab vedotin in relapsed/refractory Hodgkin lymphoma and relapsed/refractory systemic anaplastic large-cell lymphoma in the Named Patient Program: Review of the literature. Crit Rev Oncol Hematol. 2015;95(3):359-369.
- 10. Sureda A, Constans M, Iriondo A, et al. Prognostic factors affecting long-term outcome after stem cell transplantation in Hodgkin's lymphoma autografted after a first relapse. Ann Oncol. 2005;16(4):625-633.
- 11. Gibb A, Jones C, Bloor A, et al. Brentuximab vedotin in refractory CD30+ lymphomas: a bridge to allogeneic transplantation in approximately one quarter of patients treated on a Named Patient Programme at a single UK center. Haematologica. 2013;98(4):611-614.
- 12. Illidge T, Bouabdallah R, Chen R, et al. Allogeneic transplant following brentuximab vedotin in patients with relapsed or refractory Hodgkin lymphoma and systemic anaplastic large cell lymphoma. Leuk Lymphoma. 2015;56(3):703-710.
- 13. Pro B, Advani R, Brice P, et al. Four-year survival data from an ongoing pivotal phase 2 study of brentuximab vedotin in patients with relapsed or refractory systemic anaplastic large cell lymphoma. Blood. 2014;124(21 suppl): Abstract 3095.
- 14. Pro B, Advani R, Brice P, et al. Five-Year Survival Data from a Pivotal Phase 2
 Study of Brentuximab Vedotin in Patients with Relapsed or Refractory Systemic
 Anaplastic Large Cell Lymphoma. Blood. 2016;128(22):4144.
- 15. Broccoli A, Derenzini E, Pellegrini C, et al. Complete response of relapsed systemic and cutaneous anaplastic large cell lymphoma using brentuximab vedotin: 2 case reports. Clin Lymphoma Myeloma Leuk. 2013;134(4):493–495.

- 16. Monjanel H, Malphettes M, Deville L, et al. Brentuximab vedotin in heavily treated Hodgkin and anaplastic lymphoma, a single center study on 45 patients. Hematol Oncol. 2013;31(s1):268.
- 17. Lamarque M, Bossard C, Contejean A, et al. Brentuximab vedotin in refractory or relapsed peripheral T-cell lymphomas: the French named patient program experience in 56 patients. Haematologica. 2016;101(3):e103-106.
- 18. Gandolfi L, Pellegrini C, Casadei B, et al. Long-Term Responders After Brentuximab Vedotin: Single-Center Experience on Relapsed and Refractory Hodgkin Lymphoma and Anaplastic Large Cell Lymphoma Patients. Oncologist. 2016;21(12):1436-1441.
- 19. Argnani L, Zinzani PL. Reporting real-life experience with drugs in lymphoma patients. Hematol Oncol. 2016 Mar 21. doi: 10.1002/hon.2284 [Epub ahead of print]
- 20. Cheson BD, Pfistner B, Juweid ME, et al. Revised response criteria for malignant lymphoma. J Clin Oncol. 2007;25(5): 579-586.
- 21. Beaven AW, Diehl LF. Peripheral T-cell lymphoma, NOS, and anaplastic large cell lymphoma. Hematology Am Soc Hematol Educ Program. 2015;2015:550-558.

Table 1: Patient demographics and characteristics at baseline.

Total population, N	40
ALK+	22
ALK-	18
Median age, years (range)	51.4 (22.6-80.7)
Median time from diagnosis-BV, years	2 (1-16)
(range)	
Male, N (%)	28 (70.0)
Stage, N (%)	
- 1/11	9 (22.5)
- III	5 (12.5)
- IV	26 (65.0)
Systemic symptoms, N (%)	11 (27.5)
- Refractory to most recent therapy, N (%)	24 (60.0)
- Refractory to first line therapy, N (%)	25 (62.5)
Median number of previous therapies (range)	2 (2-10)
Prior autologous stem cell transplant, N (%)	13 (32.5)
Prior radiotherapy, N (%)	8 (20)

ALK: anaplastic lymphoma kinase; BV: brentuximab vedotin; ECOG: Eastern Cooperative Oncology Group.

Figure Legends

Figure 1: Overall survival.

Figure 2: Progression free survival.

Figure 3: Disease free survival.





